[#BloombergLP-bdlt-MonthOfYear-Enum]
= xref:BloombergLP.adoc[BloombergLP]::xref:BloombergLP/bdlt.adoc[bdlt]::xref:BloombergLP/bdlt/MonthOfYear.adoc[MonthOfYear]::Enum
:relfileprefix: ../../../
:mrdocs:

Define the list of month&hyphen;of&hyphen;year values&period;

== Synopsis

Declared in `&lt;bdlt&lowbar;monthofyear&period;h&gt;`

[source,cpp,subs="verbatim,replacements,macros,-callouts"]
----
enum Enum;
----

== Members

[cols="1,4"]
|===
| Name| Description
| `e&lowbar;JAN` 
| January&period;
| `e&lowbar;JANUARY` 
| Alias for `e&lowbar;JAN`&period;
| `e&lowbar;FEB` 
| February&period;
| `e&lowbar;FEBRUARY` 
| Alias for `e&lowbar;FEB`&period;
| `e&lowbar;MAR` 
| March&period;
| `e&lowbar;MARCH` 
| Alias for `e&lowbar;MAR`&period;
| `e&lowbar;APR` 
| April&period;
| `e&lowbar;APRIL` 
| Alias for `e&lowbar;APR`&period;
| `e&lowbar;MAY` 
| May&period;
| `e&lowbar;JUN` 
| June&period;
| `e&lowbar;JUNE` 
| Alias for `e&lowbar;JUN`&period;
| `e&lowbar;JUL` 
| July&period;
| `e&lowbar;JULY` 
| Alias for `e&lowbar;JUL`&period;
| `e&lowbar;AUG` 
| August&period;
| `e&lowbar;AUGUST` 
| Alias for `e&lowbar;AUG`&period;
| `e&lowbar;SEP` 
| September&period;
| `e&lowbar;SEPTEMBER` 
| Alias for `e&lowbar;SEP`&period;
| `e&lowbar;OCT` 
| October&period;
| `e&lowbar;OCTOBER` 
| Alias for `e&lowbar;OCT`&period;
| `e&lowbar;NOV` 
| November&period;
| `e&lowbar;NOVEMBER` 
| Alias for `e&lowbar;NOV`&period;
| `e&lowbar;DEC` 
| December&period;
| `e&lowbar;DECEMBER` 
| Alias for `e&lowbar;DEC`&period;
| `JAN`  [.small]#[deprecated]#
| Deprecated three&hyphen;letter aliases for the month enumerators&period;
| `FEB`  [.small]#[deprecated]#
| Deprecated three&hyphen;letter aliases for the month enumerators&period;
| `MAR`  [.small]#[deprecated]#
| Deprecated three&hyphen;letter aliases for the month enumerators&period;
| `APR`  [.small]#[deprecated]#
| Deprecated three&hyphen;letter aliases for the month enumerators&period;
| `MAY`  [.small]#[deprecated]#
| Deprecated three&hyphen;letter aliases for the month enumerators&period;
| `JUN`  [.small]#[deprecated]#
| Deprecated three&hyphen;letter aliases for the month enumerators&period;
| `JUL`  [.small]#[deprecated]#
| Deprecated three&hyphen;letter aliases for the month enumerators&period;
| `AUG`  [.small]#[deprecated]#
| Deprecated three&hyphen;letter aliases for the month enumerators&period;
| `SEP`  [.small]#[deprecated]#
| Deprecated three&hyphen;letter aliases for the month enumerators&period;
| `OCT`  [.small]#[deprecated]#
| Deprecated three&hyphen;letter aliases for the month enumerators&period;
| `NOV`  [.small]#[deprecated]#
| Deprecated three&hyphen;letter aliases for the month enumerators&period;
| `DEC`  [.small]#[deprecated]#
| Deprecated three&hyphen;letter aliases for the month enumerators&period;
| `JANUARY`  [.small]#[deprecated]#
| Deprecated three&hyphen;letter aliases for the month enumerators&period;
| `FEBRUARY`  [.small]#[deprecated]#
| Deprecated three&hyphen;letter aliases for the month enumerators&period;
| `MARCH`  [.small]#[deprecated]#
| Deprecated three&hyphen;letter aliases for the month enumerators&period;
| `APRIL`  [.small]#[deprecated]#
| Deprecated three&hyphen;letter aliases for the month enumerators&period;
| `JUNE`  [.small]#[deprecated]#
| Deprecated three&hyphen;letter aliases for the month enumerators&period;
| `JULY`  [.small]#[deprecated]#
| Deprecated three&hyphen;letter aliases for the month enumerators&period;
| `AUGUST`  [.small]#[deprecated]#
| Deprecated three&hyphen;letter aliases for the month enumerators&period;
| `SEPTEMBER`  [.small]#[deprecated]#
| Deprecated three&hyphen;letter aliases for the month enumerators&period;
| `OCTOBER`  [.small]#[deprecated]#
| Deprecated three&hyphen;letter aliases for the month enumerators&period;
| `NOVEMBER`  [.small]#[deprecated]#
| Deprecated three&hyphen;letter aliases for the month enumerators&period;
| `DECEMBER`  [.small]#[deprecated]#
| Deprecated three&hyphen;letter aliases for the month enumerators&period;
| `BDET&lowbar;FEB`  [.small]#[deprecated]#
| Deprecated `BDET&lowbar;&ast;` aliases for month enumerators&period;
| `BDET&lowbar;DECEMBER`  [.small]#[deprecated]#
| Deprecated `BDET&lowbar;&ast;` aliases for month enumerators&period;
|===


== Non-Member Functions

[cols="1,4"]
|===
| Name| Description
| xref:BloombergLP/bdex_VersionFunctions/maxSupportedVersion-05.adoc[`&colon;&colon;BloombergLP&colon;&colon;bdex&lowbar;VersionFunctions&colon;&colon;maxSupportedVersion`]
| Return the maximum supported BDEX version for `MonthOfYear&colon;&colon;Enum`&period;
| xref:BloombergLP/bdlt/maxSupportedBdexVersion-079.adoc[`&colon;&colon;BloombergLP&colon;&colon;bdlt&colon;&colon;maxSupportedBdexVersion`]
| Return the maximum valid BDEX format version, as indicated by the specified `versionSelector`, to be passed to the `bdexStreamOut` method while streaming an object of the type `MonthOfYear&colon;&colon;Enum`&period; Note that it is highly recommended that `versionSelector` be formatted as &quot;YYYYMMDD&quot;, a date representation&period; Also note that `versionSelector` should be a _compile_&hyphen;time&hyphen;chosen value that selects a format version supported by both externalizer and unexternalizer&period; See the `bslx` package&hyphen;level documentation for more information on BDEX streaming of value&hyphen;semantic types and containers&period;
|===


[.small]#Created with https://www.mrdocs.com[MrDocs]#
